BLE Beacons for Internet-of-Things Applications: Survey, Challenges and Opportunities

Kang Eun Jeon, James She, Perm Soonsawad, and Pai Chet Ng

Abstract-While the Internet of Things (IoT) is driving a transformation of current society towards a smarter one, new areas of challenges and opportunities have also arisen to accommodate the increasing demands of the IoT developments. Low power wireless devices, undoubtedly, are the viable solution for diverse IoT use cases. Among them, Bluetooth Low Energy (BLE) beacon has emerged as one of the most promising candidates due to the ubiquitous use of Bluetooth-compatible devices, such as Androids and iPhones. However, for BLE beacons to continue penetrating the IoT ecosystem from a holistic manner, a seamless integration in interdisciplinary research is needed. This paper consolidates state-of-the-art BLE beacon starting from its application and deployment cases, hardware requirements, casing designs to software and protocol design and delivers a timely review regarding related research challenges in BLE beacons. In particular, the latest developments of cutting-edge applications, the interoperability between its protocol, the reliability of signal detection and distance estimation methods, the sustainability of its low energy, the deployment constraint are discussed to identify related research opportunities and directions.

Index Terms—Bluetooth Low Energy, BLE Beacons, Internet of Things (IoT).

I. INTRODUCTION

HE integration of recently emerging technologies, namely, low power wireless technology and mobile computing, has led to the development of the Internet of things (IoT) [1], [2], which realizes the ubiquitous computing concept [3] laid down by the late Mark Weiser [4]. Recent advancement of low power wireless technologies, such as Radio-frequency Identification (RFID), ZigBee, 6LoPan, Bluetooth Low Energy (BLE), etc., has revolutionized wireless communication between devices. Such technologies have removed the hassles caused by traditional wired communication and allowed dynamic data transmission between devices over the air. The maturity of RFID technology had inspired [5] encapsulation of RFID, embedded sensing system, and ad-hoc networking for a largescale network of smart objects for the IoT [6], [7]. ZigBee, on the other hand, has been widely used for wireless home automation network [8], [9], and commercial applications development [10]. Among them, this paper focuses on surveying the latest development of BLE and its corresponding influence on development of IoT technologies and applications.

As a successor to its previous version – Bluetooth Classic, where its primary aim is in providing effective high data rate for audio and data streaming applications, BLE, on the other hand,

K. E. Jeon, J. She, P. Soonsawad, and P. C. Ng was with HKUST-NIE Social Media Lab, the Department of Electrical and Computer Engineering, The Hong Kong University of Science and Technology, Hong Kong, e-mail: {kejeon, eejames, psoonsawad, pcng}@ust.hk.



Fig. 1. Real-life use cases of BLE beacons in IoT applications: (a) interactive exhibition in Guggen-heim museum, New York; (b) interactive content distribution system by CyPhy Media Ltd., Hong Kong; (c) vending machine with location-based digital payment service by LINE Corp., Japan; (d) pull-notification based advertisement in a retail store by LINE Corp. at Uniqlo, Japan; (e) pull-notification based advertisement and promotion by LINE Corp. at BTS station, Thailand; (f) indoor navigation system with augmented reality in Gatwick Airport, England.

evolved to be an energy efficient low data rate device suitable for power constrained IoT applications [11], [12]. Since BLE unifies the advantages of both unmanned power constrained IoT applications and Bluetooth-enabled smart devices, it is witnessing an increasing number of adoption; BLE beacon is one of its most promising subset. The ease of integration between off-the-shelf BLE beacons and smartphone particularly, has empowered diverse IoT use cases especially in the emerging unmanned IoT applications, creating less efforts for human to do any task [2], [13]. The reinforcement of BLE beacon infrastructure with priority on IoT development has gained much research interests from both academia and industrial sectors, leading to limitless possibilities for IoT innovation to accommodate the needs of heterogeneous ecosystem. BLE

Manuscript received Aug 4, 2017; revised Nov 20, 2017.

beacons has been employed in a wide range of emerging IoT innovations. namely, improving shopping experience of users [14], guiding in museums [15], indoor localization and tracking [16], helping blind or vulnerabilities people [17]–[20], saving energy for smart office [21], [22], managing smart home [23] and warehouse [24], locating BLE devices with beacons using fingerprinting [25] and so on. Furthermore, increasing number of BLE beacon infrastructures are being deployed for commercial uses; some of these real-life deployment examples are shown in Fig. 1. [26] provided a forecast of a deployment of 19 billion Bluetooth devices over the next three years. Unquestionably, the adoption of BLE beacons in the IoT ecosystem will soon embark enormous research opportunities with its low power operation capabilities [27].

Recognizing the promising features of BLE beacons for the IoT development, this paper surveys the holistic development of BLE beacons for IoT applications and developments. The article is centered around presenting different applications and features of beacons including its protocol design, characteristics of Bluetooth signal, hardware components, casing designs, and software development in realizing an inter-operable, easy-todeploy and scalable beacon-based IoT solution. Denoting the key requirements of an IoT application, i.e., interoperability, detection accuracy, energy efficiency, deployment flexibility, application processing latency and system scalability, this survey further identify the relevant issues in BLE beacons and reviews its current development based on these factors. The main contributions of this paper are summarized as follows:

- an overview of BLE protocol and beacons, its applications, and related hardware and software issues;
- 2) a survey of state-of-the art research on BLE beacons;
- a review of limitations of BLE beacons and suggestion of future research directions, challenges and opportunities;

To the best of our knowledge, this article is the first attempt from academia to present a holistic overview of BLE beacon for IoT solution, considering every aspects of beacons from BLE protocol and applications to hardware design conditioned for practicality and real-life deployment.

The rest of the paper is organized as follows. Section II presents an overview of latest developments of different application leveraging BLE beacon infrastructure and further hints its possibilities and drawbacks. Section III reviews the BLE beacons in connection to its beaconing protocols and received signal strength. Section IV presents the beacon hardware from inside out, from its chipset, energy storage to its casing. Section V studies the enabling software and systems for BLE beacon and such as battery monitoring techniques, distance estimation, security features and server scalability. Section VI identifies the research challenges with respect to the three sections discussed above and Section VII concludes the paper.

II. BLE BEACON APPLICATIONS

BLE beacons have been adopted and deployed frequently over the last few years. With big industrial players, such as Google, Apple, Facebook and LINE, pushing for new standards and hardwares, BLE beacon-based services are now more accessible to both the public and developers than ever before. Riding with the tide, many interesting applications of BLE beacons have been proposed from both academic and industrial sectors. These applications include indoor localization, proximity detection and activity sensing. Following section reviews the state-of-the-art applications empowered by BLE beacons from both of the sectors and also shares our visions on potential creative applications.

A. Localization

Localization is one of the most prospected application of BLE beacons. Global positioning system (GPS) that has revolutionized outdoor localization has proven to be ineffective in indoor environment and city streets due to severe attenuation and multi-path fading effects. Wi-Fi access point based solutions has limitations due to limited number of APs and their flexibility in deployment. Namely, Wi-Fi APs are installed for signal coverage and not for localization. Other technologies such as RFID, ultra-wideband and infrared have been emplyoed. However, these devices require a dedicated reader inorder to operate. Therefore, it is hard for general public to fully utilize such service. BLE beacon-based solution have a decided advantage over currently existing solution due to its low-production cost, easy to deploy, and easily accessible by the users. Feasibility of BLE beacon based indoor localization system has been extensively investigated [25]. In their investigation, authors discuss the accuracy that can be achieved with given configuration for deployment and operation parameter, namely denseness of deployment, advertising interval, transmission power etc. In their investigation, they had setup 19 beacons in an office area and have achieved < 2.6 m error 95% of the time when 1 beacon was deployed every 30 m², out-performing <8.5 m error when using existing Wi-Fi network.

[28] presents a stigmergic approach for indoor localization that strongly leverages RSSI information provided by static anchor nodes. Such method alleviates severe attenuation that any BLE signal may suffer in crowded areas. Recently such systems have been deployed for all sorts of different environments, namely museums, airports, and to help visually impaired to navigate indoor facilities. From real-life cases, Hong Kong international airport [29], Hamad international airport [30], Gatwick international airport [31], Dallas and Houston international airport [32] have deployed beacons to aid the passengers in navigating usually unfamiliar grounds. Among them, Gatwick international airport is an interesting example as they are combining augmented reality technology with BLE beacon-based localization system.

B. Proximity Detection and Interaction

Besides providing users with locational information, BLE beacons can also convey contextual information by proximity measurement to an object or area. The difference between location and proximity is often blurred and therefore confusing. In this article proximity refers to relative distance to an object where as location refers to absolute position within a given environment. This means a beacon may be attached to nonstationary objects as well. The proximity information may trigger some event allowing seamless interaction between a user and an object. Technologies that achieves similar purpose, such as QR code and NFC exists. However, QR code needs to be installed or printed largely to reach larger number of audiences; however, design of QR code is not the most aesthetic. NFC has very short interaction distance 10 - 20 cm, which requires users to approach the media before interacting. On the other hand, BLE beacon can address both of these concerns.

BLE beacon-based proximity detection systems have already been deployed and demonstrated in real life to send effective notifications that strongly leverages on user context/location. [33] has implemented proximity detection based tour and navigation system. The system provided time table of nearby bus stop and distance to nearby subway station to the tourists. [34] has demonstrated the interactive system for art galleries, which has outperformed the conventional QR code's engagement conversion rate and time. Similarly, Estimote has implemented a BLE-beacon based system in a museum to provide detailed information about an artwork to nearby users [35]. The system employs a pull mechanism, where the information is only provided at an request. On the other hand, [36] demonstrates use case of push promotion and location advertising. In this example, beacon network consisting 1000 nodes was deployed across Hong Kong.

From the industrial parties, the Physical Web, introduced by Google Inc., provided an open way to interact smoothly and rapidly with physical objects without installing any mobile application by embedding compressed URLs in the advertisement packets. These objects are enabled with this feature by deploying Eddystone protocol employing beacon near them. When a user pass within the distance range of beacons, the services on the device including nearby notifications or Google Chrome will receive signals and transmit to the proxy before receiving URLs back and show on user smart phone/device [37]. There are three main benefits of using Physical Web with beacons. Firstly, mobile users can interact with everything around easily and quickly without downloading any mobile application at first. Secondly, they can see what is advantageous around by viewing webpages linked with the space surrounding them. Thirdly, when everything in the vicinity can transmit data and advantages, a whole new experience will occur [38].

Apple, leveraging on its iBeacon standard, has been implementing proximity based services such as continuity features [39]. Most well known, and daily used feature is AirDrop. When an iOS device is looking for other devices, it is basically scanning for iBeacon signals from other iPhones and MacBooks. Facebook has introduced their own lineup of beacons to further enhance locality features to the users. Similar technology has been adopted by the automobile industry. [40] demonstrated installation of iBeacon system on a car for automatic transaction at toll booths, parking meters, gas station and more.

LINE corp. has also incorporated LINE beacon services in their mobile applications. Fully utilizing its messenger platform, LINE has integrated BLE beacon's ability to convey contextual information with chatbot services. This allows users to engage in short dialogues to retrieve desired information. For example, LINE beacon service deployed in a Japanese clothing brand, Uniqlo, allows users to receive information regarding nearby garments through its LINE messenger applications. More interestingly, LINE has collaborated with LG to create a new application, HomeChat [41] that integrates BLE beacon, chatbot and home automation technologies. This service allows user to receive status notification of their home appliances and also make simple commands such as to change the mode of their air conditioner.

C. Activity Sensing

In previous examples BLE beacons have been mainly used to provide more user-aware services by knowing their location and context through beacons. However, in the examples shown below, the information conveyed by the beacons were reversely used to help better identify the activities of the users. [42] has used BLE beacons to detect fine-grained location and movement to better identify the activity of the users with help of gesture detection technology of smart wearable devices. Knowing the user's micro-location helps to narrow down the list of possible gestures/ actions users may take. Consequently, the authors claim they have significantly reduced active sensing time up to 92.9%. Similarly, [43] has implemented a system to help to keep track of senior citizen's activity information. The system requires senior citizens to wear BLE beacon tag equipped with an accelerometer. BLE beacon signals scanned by pre-deployed fixed scanners helps to identify micro-location of the user, whereas built-in accelerometer helps to identify simple activities such as sit, stand and walk.

D. Future Applications

The section has reviewed three very distinct use cases of BLE beacons: localization, proximity detection and activity sensing. In most of the examples BLE beacons were deployed in static location. However, it would be interesting to see more application of BLE beacons on moving objects, such as cars, trains, bicycles, and humans. This may require a study on reliability of BLE beacons for mobile objects and also more study related to activity sensing. On the other hand, with machine learning on the rise, collecting user information is of paramount importance. Inspired by the use of chatbot, we believe providing contextual and locational information of the user through both localization and proximity detection open up a new paradigm of study, where machine learning will incorporate user information to make better engagements and therefore a better service.

III. BLE PROTOCOL AND RF SIGNAL CHARACTERISTICS

In order to understand BLE beacon and their role in IoT thoroughly, this section first provides an overview on the development of Bluetooth technology, and then introduces two popular BLE beacon profiles on the market. BLE protocol is a foundation upon from which these low power wireless devices are built upon. Therefore, this section elaborates on the building block of BLE particularly the protocol related to BLE beacon and its signal characteristics. Firstly, BLE protocol design and its working mechanism are introduced, followed by currently existing industrial BLE profiles: iBeacon and Eddystone. The

characteristics of Bluetooth signals, more specifically, the received signal strength (RSS), which can be measured by any Bluetooth-compatible receivers are investigated, especially their behaviors in a dense beacon environment.

A. From Classic Bluetooth to Bluetooth Low Energy

Bluetooth technology, governed by Bluetooth SIG, has been a well-defined wireless standard for short-range communication for over a decade. Initially, Bluetooth was designed to be an alternative to wired communication between devices such that to provide greater mobility for the devices' communication within the range defined by the Bluetooth signal. For example, replacing the wired mouse with Bluetooth mouse. Obviously, the determinant factor which guarantee the past success of Bluetooth is its reliability in providing hassles-free communication between two devices, and the power feature is not the top consideration for Bluetooth technology. The story changes when the IoT devices are knocking on the door, demanding for a better and lower power communication technology to empower their further development. The demands of IoT devices have driven the design of low power communication technologies, such as RFID and ZigBee described previously.

Similarly, such trends also drove Bluetooth SIG to invent their first low power version of Bluetooth, known as BLE. Note that BLE is backward incompatible with classic Bluetooth, and it is designed for IoT devices in mind rather than for shortrange devices' communication. Having said that, BLE trade-off the high speed and high data rate features in classic Bluetooth to minimize the power consumption. Table. I summarizes the key differences between classic Bluetooth and BLE. Apart from these differences, both technologies are operating on the same license-free 2.4 GHz ISM spectrum band, and the maximum range their signal can reach is determined by their transmit power. To ensure the coexistence of both technologies, Bluetooth SIG has introduced Bluetooth Smart Ready which is able to support both type of Bluetooth simultaneously. Bluetooth Smart Ready is normally found in the devices with higher computational capabilities such as smartphone and desktop. The main focus of this survey paper is BLE beacon and its promising features for IoT development, for the detail descriptions about Bluetooth Smart Ready and other roles of BLE (i.e., peripheral, central, er and observer), one can always refer to [44], [45]. The main differences between classic Bluetooth and BLE can be summarized:

- Two protocols serve different purposes and applications. Classic Bluetooth is tailored for multimedia streaming application. Whereas BLE is aiming for IoT applications where short sensor data need to be broadcast frequently.
- Two protocols leverage different wireless communication methods. As mentioned previously, Classic Bluetooth is for streaming, consequently requiring pairing between central and peripheral devices. In BLE, such operation is not necessary.
- Following the previous point, classic Bluetooth is a oneto-one communication and BLE is one-to-many communication, where the one is a BLE beacon device.

TABLE I Classic Bluetooth versus BLE

Feature	Classic Bluetooth	BLE
Symbol rate	1-3 Mbps	1 Mbps
Power consumption	1 (normalized)	0.01 - 0.5
Throughput	0.7-2.1 Mbps	305 kbps
Connection Latency	100+ ms	<6 ms
Channels	79	40
Channel Bandwidth	1 MHz	2 MHz
Peak Current	<30 mA	<15 mA

B. BLE Protocol and Profiles

As shown in Table. I, BLE divides its 2.4 GHz ISM spectrum band into 40 channels, with 3 channels (i.e., Channel 37 (2.42 GHz), 38 (2.426 GHz) and 39 (2.48 GHz)) dedicated for advertisement purpose and the rest for data exchange. The wide spacing of the advertisement channels minimize the WiFi signals operated on the same ISM band. BLE device which only responsible for advertising via Channel 37-39 is commonly known as beacon. Beacon is connectionless and only uses advertisement channels to broadcast their signals periodically. The beauty of this mechanism is that no device pairing is required to receive the signals advertised by the beacon. This advertising signal generally contains a small data payload (known as advertising Protocol data Unit (PDU) most of the time) which may include the packet header, MAC address, device's unique identifier, and a small headroom for manufacturer specific data. Both Apple and Google manipulate this small chunk encapsulated in the advertising PDU and introduce their popular beacon profiles, iBeacon [46] and Eddystone [47], respectively.

1) iBeacon by Apple: iBeacon is a popular BLE profile that was introduced by Apple Inc. at their annual Apple Worldwide Developers Conference 2013 (WWDC) [48]. Such movement of Apple has drawn a lot of attentions from both industrial and academia players, particularly regarding the possible applications that they can develop on top of this small beacon, which claims to operate for months or even years on a coin-cell battery. Such low power consumption feature is enabled with the small data size of the advertising PDU. Fig. 2 (a) shows the advertising PDU of iBeacon, which take up total 46 bytes in length [49]. Such packet structure not only enabled convenient identification of individual beacon devices, but also provided the industry with a universal standard for application development. Moreover, ever since then, many interesting location-based and proximity-based applications have been developed [16].

2) Eddystone by Google: On the other hand, Google launched their open source BLE profile, Eddystone [50], to compete with Apple's proximity marketing campaign. The launching of Eddystone has further impact the development of IoT especially with the introduction of Physical Web [51]. Different from the proprietary iBeacon, Eddystone allows seamless interaction with existing Chrome browser installed in any operating system. That is to say, that Eddystone allows more flexibility in contextual content development rather than

(a)	Adv PDU			Payload defined by iBeacon Standard								
1 byte	4 bytes	2 bytes	6 bytes	9 bytes 16 bytes		:	2 bytes	2 bytes	1 byte			
Preamble	Access Address	Header	MAC	iBeacon Prefix Universally Unique Identifier (UUID)		JID)	Major	Minor	Tx Power			
(b)	Adv PD	U		Payload defined by Eddystone Standard				ard				
1 byte	4 bytes	2 bytes	6 bytes	1 byte		1 byte	16 bytes		2 bytes			
					Frame Typ	e Ranging	UID			Reserv	e	
Preamble Access Address Header		eader MAC	URL	1 byte	1 byte		18 bytes					
	Header			Frame Typ	e Ranging	URL						
			TINA	1 byte	1 byte	2 bytes	2 bytes	5	4 bytes	4 bytes		
			Fra	Frame Typ	e TLM Version	Battery Level	Temperat	ure A	DV_CNT	SEC_CNT		

Fig. 2. Advertising PDU of (a) iBeacon and (b) Eddystone.

to build a completely independent mobile application to interact with the deployed beacon. For further comparison between iBeacon and Eddystone, one can refer to the summary provided by [52]. In general, Eddystone allows developers to switch between URL and TLM frame, as shown in Fig. 2 (b). The working principle of a URL frame is similar to the conventional QR code, whereas TLM frame allows developer to provide additional data regarding the deployed beacon. All the technical details regarding Eddystone protocol is available in Google GitHub [53].

3) Manufacturer specific custom profiles: Beside iBeacon and Eddystone, BLE protocol is flexible enough to allow manufacturers to configure customized BLE profiles for specific usage. Manufactures can add extra information they need to the beacons or change the offset of bytes for storing information; namely, battery voltage level measurements to facilitate timely management, sensor measurements for data collection, and authentication keys for better security measures. However, the application side needs to be re-designed for retrieving the correct data from the customized beacon packets. Furthermore, these profiles may evolve to become a profile by incorporating dynamic packet structure and information. Such design may be used to provide services that are more sophisticated and ultimately open up new research opportunities.

C. Received Signal Strength and Coverage Distance

One parameter of interest from a beacon, regardless of their BLE profile, is RSS [54], [55], a measurement in dBm that describes the power received at the receiving end with respect to the transmit power. The maximum range of a beacon signal of Bluetooth 4.0 is known to be 150 m; such coverage is only obtained in an open environment where line-of-sight between a transmitter and a receiver is unobstructed. Since the signal decays along its propagation path, according to the inverse square law, the received signal power P_r is inversely proportional to the square of the distance, i.e., $P_r \propto \frac{1}{d^2}$. In reality, the signal often decays much faster due to unavoidable environmental factors. To cater for the various loss factors, the relationship between the received signal power and the distance can be further defined to $P_r \propto \frac{1}{d^{\alpha}}$, where α is the

loss exponent. Typically, RSS is measured in dBm scale (i.e., $RSS = 10 \log(\frac{P_r}{1mW})$). The relationship between the RSS and distance is hence $RSS \propto -\alpha \log(d)$. That is, in logarithmic scale, the linear relationship between RSS and distance can be formulated as below,

$$RSS = -\alpha \log(d) + K \tag{1}$$

where $-\alpha$ is the loss exponent, K is the constant that intersect at RSS axis, and d is the distance measured in meter. Note that the above equation is a general path loss model which can be applied for different scenarios, in which each scenario has their own loss exponent. [56] provides a list of possible loss exponent for different scenarios.

In fact, there are no differences regarding the signal coverage of classic Bluetooth and BLE if both are configured to have same transmit power. Fig. 3 compares theoretical distance and measured distance for a beacon with different transmit power ranging from -30 dBm to 4 dBm. Theoretical distance was provided by the manufacturer, Estimote [57], and the measured distance was collected using an off-the-shelf Android smartphone. It can be observed that weaker transmit power reduces the range of signal coverage. In addition, it can be seen that the measured range deviates from theoretical range. The signal fluctuation has resulted error in theoretical distance



Fig. 3. Distance versus transmit power.

estimation that is purely based on RSS value. Prior research also had concluded that the distance estimation based on RSS is unreliable [58], [59]. Such situation gets even severe when multiple beacons are present within a small area. Following the above observation, the next subsection examines the signals behaviors in a dense environment.

D. Beacon Signals in Dense Environment

Since BLE has reduced the number of channels (total 79 channels in classic Bluetooth) to 40 and each channels are equally spaced at 2 MHz [60], such strategical arrangement in the 2.4 GHz ISM band has prevented it from overlapping with the common Wi-Fi channel. However, consider an environment with 10 beacons placed randomly, smartphone might unable to see all the beacons' signals within a short scanning period, a time window during which the smartphone listens to BLE signal nearby. Fig. 4 (a) shows the RSS variation from each of the 10 beacons, it is observed that the RSS detected are varied across time even though each of the beacons are placed in a fixed location (i.e. the beacons are static and never move during the time of experiment). While Fig. 4 (b) shows each of the beacons required less than 1s to be detected under good condition, however, under worst condition the detection time can take more than 5s. Signal propagation (e.g. multipath fading, shadowing, fading etc.) and environmental factors (e.g. movement of people in the laboratory and room temperatures) are the causes for the phenomenon of high RSS variation and detection time variation observed above, namely from beacon B2. Out of the total 751 scans, there were 12 scans that detected 0 signal, and only 1 scan was able to capture all the 10 signals.



Fig. 4. Variation in BLE Beacon signal characteristics in a dense environment: (a) the RSS; (b) the time to detect beacon signal.

IV. BLE BEACON HARDWARE

In-depth knowledge of different hardware components of a BLE beacon is crucial in constructing a physical layer capable of providing reliable and scalable service. An illustration of the components is shown in Fig. 5. A comprehensive review of



Fig. 5. An illustration of generic hardware components of BLE beacon.

available hardware options for BLE chipset, energy storages, and casings are made. Furthermore, strengths and drawbacks of the options are discussed to provide better insights.

A. Power Consumption Characteristics of BLE Beacon

In designing a BLE beacon, it is important to maximize and estimate its battery life. Maximizing the battery life will prolong the use of an infrastructure, making it more manageable and affordable. On the other hand, precise estimation of the battery life will allow timely battery replacement of beacons, utilizing the available energy resources to their limits. In order to achieve this, detailed study on beacon's power consumption characteristics is required. In this subsection, the power usage of a off-the-shelf BLE beacon is carefully analyzed. The results presented in this section is referred from the studies conducted by the authors previously [61]. During their study, CC2451, Bluetooth chipset manufactured by Texas Instruments Inc. is used. This specific chipset is chosen as a reference as it is one of the most popular BLE ICs; market share of this chipset can be seen in the later subsection, Fig. 9.

Fig. 6 shows the different states of a beacon device, where t_T is the advertising interval of the beacon, t_p is the time during which the beacon wakes up to broadcast its advertisement packet, and t_i is the time between each advertising event, during which the beacon stays idle to save energy. Furthermore, the diagram includes the initialization stage, where a considerable amount of energy is drawn. This initialization stage only occurs once during the system boot up, unless the system rebooted itself due to some fatal error. Fig. 7 presents current consumption during advertising event in more detail. An advertising event is largely divided into three different states and more specifically nine states. These states and corresponding



Fig. 6. Electrical characteristics of BLE beacon including initialization state.

current draw is shown in Table II. Since during the idle state, the device draws constant current, and initialization stage in only executed once, we are more interested in studying the current consumption characteristics exhibited in advertising event.

Average current draw during an advertising event, I_p , can be found by taking the average of different states over the time duration of advertising event. Average current consumption during an idle state can be found simply through measurement with an ammeter as it draws current steadily. By knowing these two parameters, one can calculate average current draw by taking the weighted sum with respect to the advertising interval as shown below:

$$I(t_T | t_p, I_p, I_i) = \frac{t_p I_p + t_i I_i}{t_T}$$
(2)

where I is the average current draw at an advertising interval, t_T , with advertising event duration, t_p , average current drawn during the event, I_p and current drawn during idle state, I_i . The average current draw can help us to estimate battery life of a beacon at given advertising interval, which is a crucial parameter to consider during deployment and management. Although this method has been used in industries due to its simplicity, this method assumes constant current draw even though it is a pulse draw. In order to make a better prediction, it may be required to consider different battery models that takes battery recovery effect into consideration [62], [63].



Fig. 7. Current draw of BLE beacon during advertisement event.

TABLE II Electrical characteristics of BLE beacon during broadcasting event divided into different states.

State Number	Description	t (µs)	V (mV)	I (mA)
State 1	Wake-up	480	69.12	6.91
State 2	Pre-processing	225	85.04	8.50
State 3	Pre-Rx	160	114.20	11.42
State 4	Rx	395	184.80	18.48
State 5	Rx-to-Tx	90	89.49	8.95
State 6	Tx	130	187.60	18.76
State 7	Tx-to-Rx	155	80.96	8.10
State 8	Post-processing	1070	85.03	8.50
State 9	Pre-sleep	195	47.08	4.71

B. Options for BLE Chipset

Bluetooth chipsets are currently produced by a number of different companies, namely Texas Instruments, Nordic Semiconductors, Dialog Semiconductors, and Cypress. It is commonly known that TI provides excellent reference designs and sample codes to aid developers in getting started with their project, whereas Nordic Semiconductors have very energy efficient chipsets that will help to prolong the battery life. Cypress, on the other hand, is a leading company in designing integrated chipset for low powered devices, providing many power management ICs that can operate at a cost of few hundred pico current. When choosing a processor one may inspect in terms of 4 major aspects: power consumption, flash and RAM capacity, and internal voltage regulator. Computational ability has been excluded as most BLE chipsets have converged to the ARM Cotex M0 processor.

Table III shows a comparison of representative BLE chipsets from the aforementioned manufacturers with respect to its current draw and Bluetooth version. In many cases, current draw from the radio is considered most important as CPU intensive operation is barely found in a BLE chipset. The current draw values can be mainly found on in its datasheet. However, it is important to check the usage of a regulator, as this may reduce the current draw by a couple of milli-amperes.

For nRF51 series from Nordic Semiconductors, flash storage usually comes in 2 variants: 128 and 256 KB. Generally, 128 KB is sufficient to implement basic functionalities of BLE beacon and simple features. However, if the device is running more complicated codes and calculation or require extra storage space for logging purposes, one may find 128 KB too small. Especially if one wants to develop Device Firmware Update (DFU), where a firmware of a beacon is programmed over the air with only a smartphone and mobile application, a larger flash storage is a must. Regarding RAM capacity, we have not faced much problems and have been contempt with 16 KB variants. Unless one is attempting to implement RAM retention technique for slightly better power consumption, larger RAM does not help the development or performance from our experience.

Most beacons are equipped with internal voltage regulator, so as to allow wide range of input voltage and reduce the number of extra components required for circuit manufacturing. However, its convenience may come at a cost of its efficiency. It is a general knowledge that at higher voltage the current draw will be less. However, this is not always true as many of the beacon chipsets employ low-dropout voltage for its simplicity which is energy inefficient at high voltages. To mitigate this, some of the chipsets from Nordic semiconductors have DC-DC regulator that helps to reduce current draw at higher voltage. However, this feature has known to cause instability in the chipset in 2nd revision hardware. Such incidents may motivate the developers to dedicate an efficient voltage regulator to extend battery life and avoid complications.

TABLE III A Comparison of representative BLE Chipsets from Aforementioned Manufacturers

BLE Chipset	Supported Version	Current
CC2541	Single Mode BLE v4.0	18.2 - 14.3 mA
nRF51822	Single Mode BLE v4.1	9.7 mA
PSoC 4 BLE	Single Mode BLE v4.1	15.6 mA

C. Energy Storage

There are various means to store energy, namely, disposable batteries, rechargeable batteries and supercapacitors. Many of the beacons currently available in the market employs a type of disposable lithium ion batteries; Estimote and Kontakt.io beacons both employ lithium manganese dioxide batteries for their affordable price, thermal stability, and non-toxic properties. In the following section, different means of energy storages that could be used for beacons will be reviewed and discussed.

Many BLE beacon devices prefer to use coin cell batteries due to their low-profile form factor while being able to deliver sufficient amount of power. As a proof of this, almost all of major beacon manufacturing companies uses lithium coin batteries denoted by CR or BR. However, both empirically and theoretically, these coin cell batteries have proven to last only for short period of time and thus requiring a frequent battery replacement; the Table III shows the theoretical life span of a beacon at 800 ms advertising interval, an interval often used by BLE beacon manufacturers.

Consequently, some manufacturers have employed a larger size alkaline battery, such AA or AAA batteries to extend the life span of the beacon device. However, such life span comes at a cost of larger casing footprint and heavy weight. For example, Sensoro Pro beacons from Sensoro are equipped with four AA batteries and claims to last 5-6 times longer than ordinary beacons equipped with CR2477, which has capacity of 1000 mAh. Beacons from TheBeacons uses 2 AA alkaline batteries with capacity of 2600 mAh. However, such increase in size and weight of a beacon ultimately undermines the very advantage of BLE beacon devices, convenience of deployment. BLE beacon is considered very scalable not only due to its minimalistic protocol but also because it is very easy to deploy and use. Conventional beacon usually weights around 20 - 30 g, making the deployment procedure as easy as just attaching the beacon on a wall with a simple sticky tape available in any hardware store. However, use of larger batteries usually undermines this unique advantage.

TABLE IV THEORETICAL BATTERY LIFE CALCULATION

Model	Capacity (mAh)	Size (mm x mm)	Life Time (days)
CR2477	1000	24 x 7.7	640
CR2450	620	24 x 5.0	397
CR3032	500	30 x 3.2	320
CR2032	320	20 x 3.2	205

D. Energy Harvesting Capability for BLE Beacons

To mitigate the battery issue of BLE beacon, some manufacturers have attempted to design an energy harvesting BLE beacons equipped with solar panels to sustain itself. Energy harvesting wireless sensor nodes has been a popular research topic [64]. Many researches were conducted to optimize energy harvesting hardware in terms of energy energy harvesting mechanisms, storage source, and charging circuitry. Such trend in energy harvesting untethered devices have affected the developments of IoT devices. [65] used kinetic energy harvesting method to harvesting energy from human movement and power a sensor node measuring human motion. [66] presented wireless sensor node system equipped with light harvesting capability with extremely small form factor of few millimeters. In addition, [67] used combination of RF and light harvesting to operate BLE beacon system with very long advertising interval of 45 s. Number of industries attempted to prototype similar device relying on ambient light harvesting methods. These products are reviewed in detail in Table V. However, energy harvesting capabilities of these devices are too low to support required advertising frequency of 1 Hz or storage capacity of the devices will forbid them from long-term operation in absence of ambient energy. Previous works on energy harvesting wireless sensor shown above focus mostly on outdoor deployments. However, many BLE beacon applications take place indoor. Hence, some of the previously mentioned energy sources may be absent or too scarce to harvest enough energy for perpetual operation of the untethered device. Therefore study of indoor energy harvesting is necessary in order to design energy-neutral BLE beacon device. Only recently, investigations on use of indoor lighting and photovoltaic cell for wireless sensor were carried out. [68] have provided comprehensive design consideration for indoor light energy harvesting wireless sensor system that employs MPPT technique and rechargeable battery. They claim that their prototype is expected to operate for 10 to 20 years without maintenance. [69] have presented different models of indoor energy harvesting sensors utilizing combination of capacitors and batteries. Based on previous works, generic system architecture of solar powered BLE beacon will consist of three main components: a photovoltaic energy harvesting module, power management unit, and BLE unit.

E. Casing for Look and Protection

Two major concerns for BLE Beacon casing are looks and protection. Fig. 9 summarizes main stream BLE beacons with respect to different casing designs, power sources and chipset manufacturers. A casing may have an aesthetically pleasing

 TABLE V

 Review of Commercially Available Energy Harvesting BLE beacons

Model Name Parameters	GCell Solar iBeacon	TheBeacon iBeacon Solar	Cypress SolarBeacon	TIDA Indoor Light Harvesting Beacon	HKUST SolarBeacon X1
	ccatt		DM . TM		
Size	123 x 61 x 25 mm	54 x 54 x 20 mm	25mm diameter x 5.5 mm	86.36 x 60.96 mm	12 x 28 x 36 mm
BLE Chipset	TI CC2541 (TI)	Unknown	Cypress CYBLE-022001-00	TI CC2541 (TI)	Nordic nRF51822 (ND)
Rechargeable energy storage	2 mF capacitor	120 mAh Li-ion battery	0.2 F supercapacitor	8 mF supercapacitor	17mAh Li-ion battery
Disposable energy storage	AA battery x 2	N/A	N/A	N/A	N/A
Minimum operating light intensity (l_m)	N/A	Unknown	100 Lux	250 Lux	250 Lux
Minimum advertising interval @ l_m	N/A	Unknown	45 s	1 s	1 s
Operation lifetime at full charge	N/A	90 days	Unknown	< 30 mins	100 hours
Remarks	Cannot operate without disposable battery	Very slow recharge under indoor lighting	Very low advertising frequency	Very small energy storage	Difficult to recharge in indoor settings

design that may even serve as a decoration. Such designs encourage venues, such as retail shops, to deploying these devices. Estimote beacon is a good example. On the other hand, the casing may be designed to be homogeneous and with the environment, making the device less noticeable. For instance, Gimbal's S21 beacons were made in white color and invisible brand name to achieve this purpose. In order to provide reliable and long-lasting service, it is vital to protect its inner circuit from water, dust and impact from potential abuses. To meet these demands, latest beacon casing designs often follow water/dust resistant standards such as the International Electrotechnical Commission's (IEC) Ingress Protection (IP) code. However, most of the off-the-shelf BLE beacons are not protected for long. Estimote's beacon is able to withstand high water pressure, but the case has to be cut to replace the battery inside. Consequently, the beacon will no longer have the water-resistant feature after its first maintenance. This is a common issue for most of the protective casings currently available in the market.

F. Casing for Installation and Deployment

The Installation casings, which are often overlooked and not incorporated into the designs, play a pivotal role in deployment and maintenance procedures. Compared to the traditional method, where a double-sided adhesive tape was used to install beacon, installation case can fix the beacon much more securely; we have experienced that adhesive tapes can be very weak for materials such as wood. Furthermore, they can provide removal mechanism to easily detach BLE beacon from its installation casing for battery replacement as shown in Fig.



Fig. 8. Different features of beacon casings: (a) aesthetic design, Facebook beacon; (b) water-resistance, Sensoro Pro beacon; (c) installation brackets, GCell G300 Universal iBeacon; (d) neck lanyard and a card holder design, Bright Beacon.

8 (c). This GCell beacon has an installation case with an installation brackets, which is meant to be drilled to a wall. Kontakt has also designed Beacon Pro with a mounting clip at the back to deploy and dismount this beacon. However, it is not easy to used these type of installation casings as they will damage the deployment location. Clearly, to achieve both solid installation without damaging the deployment facilities, an approach from a new angle is imperative.



Fig. 9. Review of commercially available BLE beacons categorized by casing, power source and chipset.



Fig. 10. Examples of beacon deployment locations: (a) HKUST campus, Hong Kong; (b) Outdoor news-stands, Hong Kong; (c) BTS Skytrain, Bangkok.

Design of installation casing may also be influenced by material and orientation of installation surface. For real-life deployment locations, CyPhy Media beacons were deployed in 3 major locations: Hong Kong University of Science and Technology (HKUST), various news-stands across Hong Kong and BTS Skytrain stations in Bangkok as shown in shown in Fig. 10. From Table VI, beacons were deployed on vertical surfaces more than horizontal ones in BKK and HK. However, in HKUST, beacons were deployed horizontally more than vertically because they may need to be hidden from line of sight, for example under a table for some furnitures. However, such method of deployment is not ideal in signal propagation as BLE signals are extremely easy to attenuate. When we were deploying beacons in Fig. 10 (c), we have attempted to install the beacon inside the aluminum column to protect it from weather and people. However, due to severe attenuation from the aluminum plates, BLE beacon signal could barely be detected. To avoid such degradation in performance, it is highly recommended to ensure nothing obstructs beacon signal. Therefore, it is most appropriate to install beacons at a height, which will ensure line-of-sight in many cases and also protect it from physical attacks. Furthermore, since BLE beacons signal attenuates with human body, as it is composed mainly of water, BLE beacon signal will attenuate more in crowded area. Placing beacons in higher location alleviates such attenuation effect.

Table VII shows materials of three major deployment locations are mostly made of metal. In Hong Kong, wood and plastic share similar number of deployed beacons after metal, and beacons were only deployed on aluminum advertising sign at BTS Skytrain columns in Bangkok. We have found that double-sided adhesive tapes are not suitable for surfaces like wood, therefore different installation method may be employed depending on deployment surfaces. Noting that, metal is the most popular deployment surface material, the potential for installation casing leveraging on magnets can be envisioned. Since beacons are very light-weighted, use of magnets and high-friction material, such as rubber pads, may work well on any metal surfaces. However, a study on performance of BLE signals near magnets need to be studied.

Case shapes and sizes may have very little or no effects on BLE signals because they are mostly made of plastic or

 TABLE VI

 CYPHY BEACONS PLANES IN 3 MAJOR LOCATIONS

Locations	Vertical	Horizontal	Slope	Total
HKUST	29 (36%)	45 (56%)	6 (8%)	80
News-stands, HK	101 (94%)	7 (6%)	0	108
BTS Skytrain, BKK	217 (100%)	0	0	217

 TABLE VII

 MATERIALS OF DEPLOYMENT LOCATIONS IN 3 MAJOR LOCATIONS

Locations	Metal	Wood	Plastic	Others
HKUST	32 (40%)	15 (19%)	15 (19%)	18 (22%)
News-stands, Hong Kong	61 (56%)	20 (19%)	26 (24%)	1 (1%)
BTS Skytrain, Bangkok	217 (100%)	0	0	0

silicon based rubber like material. Deployment environment can influence BLE signal due to refraction and obscuration. For example, beacons deployed hidden from line of sight to avoid vandalism and theft may affect the signal propagation and therefore its coverage. However, it is worth to investigate whether what type of case materials can enhance the propagation of BLE signal and protection of circuit components from external factors such as water and dust.

V. SOFTWARE AND SYSTEM FOR BLE BEACON

Although protocol and hardware developments of BLE beacon has laid a strong foundation upon which IoT applications and services can be implemented on, BLE beacon has drawbacks that arises from its inherent architecture, namely, large fluctuation in RSS and finite battery capacity. Such weaknesses makes beacon infrastructure difficult to implement and manage. Leveraging on power of big data and advanced signal processing techniques, aforementioned shortcomings can be overcomes thorough softwarization. Software and system for BLE beacon infrastructure includes battery measurement, distance estimatoin, security features and scalable server architecture and algorithms. In this section, current state of aforementioned developments are reviewed and discussed in detail.



Fig. 11. Comparison of beacons battery level measured by mobile application and actual value: (a) full voltage range; (b) working voltage range.

A. Battery Monitoring

After deploying a BLE beacon infrastructure, monitoring the battery levels and replace the battery on time is necessary for management. According to the Eddystone-TLM Advertising Packet Specification in Fig. 2 (b), the battery voltage level is built inside the TLM frame, 2 and 3 bytes offset. When a smart device interacts with an Eddystore-TLM protocol BLE beacon, it can get the battery information together with the advertising packet and extract the battery level. However, iBeacon's battery level cannot be found inside its standard advertising packet Fig. 2 (a). For iBeacon, manufacturers can add extra packet or configure unused bytes for BLE beacons to store battery information when they are being produced [70]. Therefore, a smart device can request the packet storing battery information when the beacons broadcast signal to smart devices. Kontakt.io provides beacon with iBeacon protocol and supports battery level monitoring. They are also using this method by storing the battery level information in 23 bytes. The information is presented in decimal number and need to be rephrased on the application side. Based on the methods mentioned, BLE beacons provide flexibility for users to collect battery information by interacting with them, especially for iBeacon protocol. Most of the beacon related SDKs and libraries also include function to collect their beacon's battery level such as Estimote, AltBeacon, etc.

To verify the accuracy of battery monitoring, an experiment on comparing the measured battery level and the actual battery voltage level, the results is shown in Fig. 11 (a). The beacon's battery model is CR2450 with nominal voltage of 3 V. The graph shows the measured battery percentage by mobile app in between the theoretical working voltage level (2 - 3.6 V). Battery level was measured using a mobile application installed on iOS device provided by CyPhy Media. The result shows that the measured battery level starts getting zero value when the voltage drop to under 2.7 V. Possible reasons are the beacon cannot work under 2.7 V or the RSS from the beacon cannot be detected by the smartphone. By studying the working range in this experiment showing in Fig. 11 (b), which is very similar to CR2450 voltage characteristic [71] showing that the battery monitoring method is able to provide approximate battery level information for user to reference if the beacon signal can be

detected by the smartphone based on this experiment.

B. Distance Estimation

Distance estimation is a key-enabler to many IoT applications. While RSS is the cost-efficient method for distance computation, the fluctuation of RSS affects the reliability of the final estimation result. Let's take a look at the distance estimation algorithm provided by Apple; this algorithm is available with the CoreLocation framework for iBeacon related development. We implemented a simple app using their framework and conducted an experiment to measure the distance. A total of 60 samples were taken for each 0.2 meter. The result is shown in Fig. 12. Obviously, the estimation error increases when the distance increases. In general, the estimation is reliable only up to the first 0.5 meters.

Several studies have indicated that the distance estimation can be improved by first obtaining a reliable RSS measurement. [72] introduced a RSS threshold optimization method to improve RSS for estimating distance for indoor application. [73] also concluded that the error rate of distance measurement is even higher than RSS measurement based on the experiment result. Therefore providing accurate distance information is very important for developing IoT applications. An improvement on traditional centroid localization algorithm has been made and given a significant accuracy improvement at 63% [74]. The algorithm in measures three intersecting point of the beacons region and calculates the beacon position. Another approach is to calculate a path loss index by comparing the RSS in 1 m and the target distance [75]. The algorithm gives an average of 0.4 m error for experiment within 3.5 m. Based on the discussion here, various estimation method can be used based on users' need. Therefore, third-party beacon SDK like AltBeacon, Estimote who support distance estimation use their own algorithm to measure distance of their beacons.

C. Security Features

Although extremely scalable due to its simplistic broadcasting architecture, BLE beacon infrastructure can be easily abused and attacked by unauthorized parties. Such attacks include but are not limited to physical attacks such as thievery and vandalism, but also cyber-attacks and sabotaging such



Fig. 12. Comparison between estimated distance from RSS measurements and actual distance.

as piggybacking, device spoofing, packet injection, beacon hijacking, denial of service attack, battery drainage attack, and selective frequency jamming. Piggybacking is a kind of an abuse where an unauthorized party uses the beacon infrastructure without prior consent from the infrastructure owner. Such abuse can be done because beacon advertisement packets are static and therefore can be easily recorded and remapped to any content on a different online server. Beacon spoofing is an act where an advertisement packet of a beacon is cloned to a different beacon device, thereby impersonating or "spoofing" the original beacon. Such abuse can be problematic in some cases, as beacon spoofing enables beacon infrastructure service outside its service area; in some applications, such as scavenger hunt at CES 2014 [76], 2016 [77], beacon spoofing is undesired. Packet injection is very similar to beacon spoofing, but instead of placing cloned beacon outside the service area, it is placed among the original network, disturbing its normal operation. Therefore, packet injection is considered as a type of an attack. In case of localization services, such attack may lead to critical system malfunction. Potential damages of this type of attack is well illustrated in [78]. Battery drainage attack on a single beacon, rendering it inoperable, was also demonstrated in [79].

Currently, most solutions to securing beacon infrastructure is proposed by the industries, namely geolocation validation and cloud-based token authentication. In geolocation validation approach, geolocational information of individual beacons are pre-registered on an online server. On the user mobile side, location information provided by GPS module is transmitted to the server along with detected beacon signals, thereby ensuring physical presence of the user near the detected beacons. This approach can secure beacon infrastructure from beacon spoofing attacks. However, such security framework has many loop holes. Firstly, the operation of pre-registering every beacon's geolocation information on online server is tedious and resource consuming, therefore reducing the scalability of the approach. Furthermore, since GPS readings in indoor environment in unreliable, geolocation validation would be restricted for outdoor use only. Another approach is cloudbased token authentication method. In this approach, beacons are provided with an algorithm to generate beacon ID based on some token value. This token value is the true ID of the beacon and can only be deciphered by the cloud server. However, this

13

method is implemented at the firmware level of the device, which means, once the algorithm generating the beacon ID is discovered by the attacker, the system can be abused. Secondly, such framework is difficult to deploy onto already existing infrastructure, as it requires individual beacon's firmware.

D. System Scalability

BLE beacons are normally used on the applications interacting between beacons and the edges (e.g., smartphone, smart wearable, etc.). These applications are not just measuring RSS or estimating distance, they also involve network requests to corresponding cloud servers. Therefore, study of server scalability is necessary for developing a beacon system to help optimizing the loading and improve performance for beacon application. After ranging a new beacon, the edges will send the unique beacon's identifier to servers using HTTP requests for getting related information. For example, a smart shipping system in mall, users are able to purchase items or get store coupons when they are close to the shop's beacon(s) [80]. The network requests can use both POST or GET methods, depends the application needs [81]. POST method are usually used for updating content on the server side, it will not include the request information in the URL body so it can provide a better security level. GET does not involve write-in process to the database, can reduce the loading on the server side but the request information needs to be added to the URL body means that it will cause security issue while sending some credentials.

Considering there are n beacons interacting with one user at the same time, there will be n requests sent by the user to the server for getting information. If there are m users interacting the *n* beacons at the same times, there will be $m \times n$ requests being sent to the server, and certain amount of responses will be sent back. Fig.13 shows an example of a server interacts with a beacon network with 2 beacons and there are 2 users in the region interacting with both two beacons, a total of 4 requests and responses will be generated normally in the case. When the scale of the beacon network increase, more users will interact with more beacons and the requests will increase accordingly. Therefore, deployment of server is important for beacon application systems. Amazon Elastic Compute Cloud (Amazon EC2) [82] is one of the possible choice providing good scalability and performance service that many companies such as Netflix, Adobe, etc. are using [83]. Tools like Jmeter [84] can simulate real network requests to server for testing the scalability of a server. A scalable server is expected to able to maintain good successful connection rate when requests number and packet size increase.

VI. RESEARCH CHALLENGES AND OPPORTUNITIES

At close inspection of BLE beacon technology, there is no doubt on its feasibility and suitability as IoT infrastructure: flexibility in BLE protocol allows great degree of freedom for developers, low cost hardware and ease of deployment makes the infrastructure more affordable and scalable. However, it is evident that there are still some drawbacks arising from its inherent design, such as interoperability between different BLE profiles, short battery life, security issue and so on. In this



Fig. 13. Interaction between beacons, edges and server

section, such demerits of BLE beacon are discussed and future research directions are suggested accordingly.

A. Challenges about the Protocols

In BLE beacon context, BLE profile simply defines the data structure or the format of the advertising PDU. This subsection discusses the interoperability challenge across two major profiles: iBeacon and Eddystone. Note that both iBeacon and Eddystone are incompatible. Even though some manufacturers create a beacon to support above two protocols, the problem is they can only support one protocol at a time, and developers or users need to switch between these protocols manually. At the time of this survey writing, no beacons on the market can support both protocols running concurrently at the same time. While most manufacturers incorporate the switching mechanism to support both protocols, this switching needs to perform during the development or configuration phase. Once the beacon is deployed with particular protocol, it is very hard to change the protocol on the fly.

The most challenging issue to deal with is BLE beacons only allocate a small chunk to customize their advertising PDU. It is, if impossible, quite hard to load both protocols within this short advertising PDU. Consider the market with diverse smartphones or BLE-enabled receivers. Failing to incorporate either one of the protocols might possibly mean that service providers will fail to reach to the other half or more of the users. Hence, to ensure a wide penetration, it is an urge to have a standardized protocol that can support both iBeacon and Eddystone at the same time, or at least a technique that allows the beacon to switch between both protocols seamlessly without human intervention.

In the era of IoT, it is expected that there will be manyto-many interaction within the same given region along with the deployment of multiple beacons. However, interference during the interaction process is an issue affecting smooth and interruption-free interaction in an environment with dense beacons deployment. In such an environment, chances are these beacons will interfere with one another if they are closely



Fig. 14. An illustration on RSS comparison approach for BLE beacon-based interaction system.

spaced [85]. As illustrated in Fig. 14, most interactions with beacons are based on the RSS-comparison approach [85], in which all RSS values are compared and only the strongest signal will be processed. However, this type of interaction only works sometimes. This undoubtedly create a challenge for the connected-things to participate in the interaction activities when incorrect RSS is processed. Furthermore, as different connected-things might use different technologies, this poses an application development challenge, as there are non-standardize interaction interface.

B. Challenges about the Hardware

The following section identify the IoT-related challenges in connection to energy efficiency of BLE beacons and the possible deployment constraint. In particular, three related issues are identified, that is battery, casing design and fixing. One of the major drawbacks of BLE beacons is rooted in its limited power source. As mentioned in previous section, battery life of BLE beacon powered by coin cell battery CR2032, most commonly used battery, will last less than a year. Such short life span of beacons requires periodic operations that replace and maintain the already deployed beacons. Although energyharvesting devices are well studied in fields of wireless sensor network, study of similar nature is deemed necessary also in the field of BLE beacons for several reasons. Firstly, wireless sensor network focused their research on energy harvesting in outdoor environment. However, BLE beacons are used widely in both indoor and outdoor environment, therefore study on energy harvesting devices in indoor setting is required. Furthermore, a robust design that considers several different deployment environment is strongly desired to enhance the scalability of the devices. Secondly, electrical characteristics of BLE beacon devices are quite different from those of wireless sensor devices. Therefore, different chipsets and hardware specifications of energy harvesting devices and storage devices are not optimal for beacon devices and should be tailored for this application. Consequently, the new research field comes down to studying

hardware specifications and optimizing it to application specific requirements.

With respect to energy harvesting BLE beacons, ambient light energy harvesting beacons have been investigated and prototyped by number of groups. However, in terms of other energy sources such as, thermoelectric, wind, acoustic, vibration, RF has not been fully explored. [67] has explored use of RF harvesting to power beacon and their energy neutral beacon could broadcast every 45 seconds at input power of 15 dBm at frequency bands between 2.4 GHz and 5 GHz. However, input power of 15 dBm is usually not feasible considering that most Wi-Fi router's transmission power is limited around 20 dBm. Furthermore, advertising interval of 45 seconds makes beacon device unsuitable to serve as proximity detection or localization infrastructure but more like a sensor device. Other energy sources such as thermoelectric and kinetic energy from vibration may require complicated deployment procedure. Thermoelectric generators requires a heat source and a heat sink to create a thermal gradient. Furthermore, thermal pastes that facilitates heat transfer must be applied during deployment procedures. For vibration, energy harvesting device must be attached to vibrating object and stationary object, so as to generate alternating current using the principles of electromagnetism. Such deployment procedure is quite complicated and painstaking. Such needs may connect to research on casing design that may help to facilitate these needs.

First challenge for beacon casing is to design a case according to the IEC standard 60529 or standard 250-2003 [86] that can retain its protection feature even after battery replacement. Since BLE beacons may require frequent battery replacement, it is important to research on devising a casing structure that can facilitate convenient battery replacement and still can effectively protect the inner circuitries from water and dust. In addition, attenuation caused by the casing should be kept minimal. If the case were made of metal, it would block the signal. Furthermore, it should still achieve small form factor by using different materials such as hydrophobic nanomaterial for better water-resistance [87]. Second challenge is to investigate cases that can increase efficiency for energy harvesting. Recently, there are ongoing investigation on photovoltaic modules assembled three-dimensional. Inspired by photosynthesizing plants, [88] has studied placement of photovoltaic module in a tree-shape following the Fibonacci number. [89] has demonstrated a 3D structure that uses photovoltaic modules as both reflector and absorber to enhance the efficiency. Such designs can be incorporated in to the casing to facilitate perpetual operation of light harvesting beacon in scarce ambient light environment. Third challenge, unique installation techniques and cases that are convenient to install, strong in protection and easy to replace. Furthermore, the design may incorporate the knowledge of deployment surface and employ adequate installation methods accordingly. Furthermore, recent advancement in synthetic setae [90] with the help of nanotechnology, may be incorporated to the casing design to replace adhesive-tape method.

C. Challenges about the Software and System

Since beacons are battery-powered device, it is hence important to have a software program that is able to accurate monitor the battery locally and remotely. Another challenge arises with application development is to deal with the unreliability of RSS to achieve a better distance estimation.

1) Battery Monitoring: The major challenges of monitoring battery level of BLE beacons is the monitoring frequency, as the battery information can only be retrieved when the beacons interact with user's smart devices. The battery information is collected by user smartphones and sent to the cloud server so that other infrastructure managers can access them. However, BLE technology requires user presence in close proximity of the monitored beacon. This means the battery information cannot be updated frequently if there are small user traffic near deployed infrastructure. Since most of the beacons are built with limited power source, monitoring the battery level frequently is a serious challenge in this stage. Beacon's battery information can be configured into the different bytes of the advertising packet. It may occur error when retrieving the battery level because the battery information packet offset may be different for each beacon, developers may not able to get beacon A's battery level by using the same method for getting beacon B's. Moreover, the presentation method of the battery information in the packet may also be different such as it can be presented as battery percentage, voltage level, etc. Therefore, developers should aware the information extracting method of every beacons so that the correct battery level can be obtained.

2) Distance Estimation: Considering the studies previously reviewed on distance estimation, accurate distance estimation is difficult to achieve due to unstable BLE signals. BLE beacon needs to work in an environment with multiple signal emitter for some special purpose such as indoor location service. Before, the closest signal source of most of the wireless communication technology can be identified easily as the signal source are not that close to each other like beacon infrastructure. Hence, measuring an accurate RSS to identify different beacons in dense environment is a new challenge on developing beacon related applications to make sure correct information is collected by smart devices. The future works need to stabilize the RSS so that the calculated distance will not fluctuate too much and cause measurement error. Some works have been done on the research area such as selfcorrection beacons [91] by comparing the RSS measured in 1 meter to obtain a more actuate value. People also proposed create different profiles on each device to achieve a better accuracy assumption as a beacon's RSS will vary for different devices. Since distance estimation is important for application development, more researches and studies need to be done in the future to improve the algorithm to achieve accurate distance estimation.

3) System Scalability: System scalability is important to study as most of the beacon applications will connect to server and can generate lots of network requests that need to be managed properly. In some condition, users may interact with more than one beacon at the same time. If there are n beacons, n requests and n responses will be generated by using normal approach. There are several ways to improve

server performance, by minimizing the server requests from the application. Developers may consider putting more controls on the network requests or filtering out only the useful beacon before sending the requests to server. For example, they can group all the requests together for certain period of time instead of requesting the server once they meet a beacon if the application does not request that frequently update.

4) Security Issue: In terms of security measures for BLE beacon network, previously introduced systems, such as cloudbased token authentication or geo-location validation systems, are more of precautionary systems rather than security systems, as it is capable of preventing any abuses, but fails to counteract or even detect any potential attacks. To remedy those drawbacks there has been a study on detection of physical attacks on beacons [92]. In their work, hidden Markov model (HMM) was used to estimate the probability of a beacon device being physical removed, relocated, and being cloned. With false-alarm rate of 5%, they have been able to achieve a promising results where removed/stolen beacons could be detected perfectly, and relocated or cloned beacons could be detected with around 70% accuracy. However, it is well known that HMM are computationally demanding, and therefore lays doubt on the practicality of the proposed system as real beacon infrastructure involves few hundreds or even thousands of beacon devices; in their work only 11 beacon devices were used in their experiments. Security features of BLE beacon infrastructure is in its infant stage and there are only handful of researches done in this field. At current stage, detection system using HMM and precautionary systems with beacon ID shuffling method are devised, however they are not a perfect security system. A secure system must be able to detect potential threats and attacks and respond appropriately while being able to take some form of precautionary measures. Therefore, a more scalable and computationally less intensive method of detecting attacks must be devised, and a security protocol that allows full control over the network is desired.

VII. CONCLUSION

Internet of Things has tremendous potential to change the modern lifestyle as the Internet has done in the past; and BLE beacons are prospected to play a pivotal part in realization of this new paradigm. Recognizing such potential in BLE beacon, this article reviews different aspects of BLE beacon for its usage in IoT application. Firstly, different applications leveraging BLE beacon infrastructure were introduced: localization, proximity detection and interaction, and activity sensing. After that, to form a good understanding of BLE beacon, BLE protocol and its signal characteristics, which is common to all BLE devices, were first reviewed. After that BLE beacon hardware was inspected regarding its generic electrical consumption characteristics. Furthermore, a model to estimate the current draw of a beacon with respect to its advertising interval was provided. Varying options available for chipsets, such as flash and RAM capacity, and internal voltage regulator were discussed. Different casings to protect and install beacons were also reviewed in detail. Software and system for BLE beacons with respect to battery measurement, distance estimation, security features, and server scalability reviewed in-depth

Based on the survey done on aforementioned topics, limitations of BLE beacons are identified and future research direction on individual topics were discussed. Interoperability between different BLE beacon devices and operation in dense BLE beacon environment are obstacles that must be overcame in order to make the infrastructure more robust. Study on sustainability issue and casings of beacon is necessary in order to make infrastructure management less resource consuming. In terms of software and system, algorithms to measure battery level and distance with precision is strongly desired. From the server side, a scalable server infrastructure and security protocol needs to be developed. Indeed, BLE beacon related studies and technology is still in its infancy; studies regarding its casings, deployment methods, security issue, and server scalability has been barely studied. However, its scalability that arises from small form factor and affordable hardware price and flexibility in protocol that can host numerous types of application, outweighs its current underdevelopment and makes it worthwhile to further investigate and research.

ACKNOWLEDGMENT

This work was supported by HKUST-NIE Social Media Laboratory. The authors would like to thank Ching Hong LAM for assistance with experiments. The authors would also like to thank CyPhy Media Ltd., Trans AD Solutions Co. Ltd. for providing various platforms to experiment.

REFERENCES

- K. Ashton, "That 'internet of things' thing," *RFiD J.*, vol. 22, no. 7, pp. 97–114, 2009.
- [2] L. Atzori, A. Iera, and G. Morabito, "The internet of things: A survey," *Comput. Netw.*, vol. 54, no. 15, pp. 2787–2805, 2010.
- [3] M. Friedewald and O. Raabe, "Ubiquitous computing: An overview of technology impacts," *Telematics Inform.*, vol. 28, no. 2, pp. 55–65, 2011.
- [4] M. Weiser, "Ubiquitous computing," Comput., no. 10, pp. 71-72, 1993.
- [5] T. Sánchez López, D. Ranasinghe, M. Harrison, and D. McFarlane, "Adding sense to the internet of things-an architecture framework for smart object systems," *Pers. Ubiq. Comput.*, pp. 1–18, 2011.
- [6] J. Cho, Y. Shim, T. Kwon, and Y. Choi, "Sarif: A novel framework for integrating wireless sensor and rfid networks," *IEEE Wireless Commun.*, vol. 14, no. 6, pp. 50–56, 2007.
- [7] H. Liu, M. Bolic, A. Nayak, and I. Stojmenović, "Taxonomy and challenges of the integration of rfid and wireless sensor networks," *IEEE Netw.*, vol. 22, no. 6, pp. 26–35, 2008.
- [8] K. Gill, S.-H. Yang, F. Yao, and X. Lu, "A zigbee-based home automation system," *IEEE Trans. Consum. Electron.*, vol. 55, no. 2, pp. 422–430, 2009.
- [9] C. Gomez and J. Paradells, "Wireless home automation networks: A survey of architectures and technologies," *IEEE Commun. Mag.*, vol. 48, no. 6, pp. 92–101, 2010.
- [10] A. Wheeler, "Commercial applications of wireless sensor networks using zigbee," *IEEE Commun. Mag.*, vol. 45, no. 4, pp. 70–77, 2007.
- [11] J. Gubbi, R. Buyya, S. Marusic, and M. Palaniswami, "Internet of things (iot): A vision, architectural elements, and future directions," *Future* generation comput. syst., vol. 29, no. 7, pp. 1645–1660, 2013.
- [12] D. Miorandi, S. Sicari, F. De Pellegrini, and I. Chlamtac, "Internet of things: Vision, applications and research challenges," *Ad Hoc Netw.*, vol. 10, no. 7, pp. 1497–1516, 2012.
- [13] S. Agrawal and M. L. Das, "Internet of things: A paradigm shift of future internet applications," in *Nirma University Int. Conf. Eng. (NUICONE)*, 2011, pp. 1–7.
- [14] D. Zaim and M. Bellafkih, "Bluetooth low energy (ble) based geomarketing system," in 11th Int. Conf. Intell. Syst.: Theories Appl. (SITA), Oct 2016, pp. 1–6.

- [15] S. Alletto, R. Cucchiara, G. D. Fiore, L. Mainetti, V. Mighali, L. Patrono, and G. Serra, "An indoor location-aware system for an iot-based smart museum," *IEEE Internet Things J.*, vol. 3, no. 2, pp. 244–253, April 2016.
- [16] M. S. Gast, Building applications with iBeacon: Proximity and location services with bluetooth low energy. O'Reilly Media, 2014.
- [17] S. A. Cheraghi, V. Namboodiri, and L. Walker, "Guidebeacon: Beaconbased indoor wayfinding for the blind, visually impaired, and disoriented," in 2017 IEEE Int. Conf. Pervasive Comput. Commun. (PerCom), 2017, pp. 121–130.
- [18] D. Kendrick, "Lowviz guide: Indoor navigation for people who are blind or visually impaired," 2015.
- [19] A. J. Ruffa, A. Stevens, N. Woodward, and T. Zonfrelli, "Assessing ibeacons as an assistive tool for blind people in denmark," *Worcester Polytechnic Institute, Interactive Qualifying Project E-project-050115-*131140, 2015.
- [20] H. J. Tay, J. Tan, and P. Narasimhan, "A survey of security vulnerabilities in bluetooth low energy beacons," 2016.
- [21] M. Choi, W.-K. Park, and I. Lee, "Smart office energy management system using bluetooth low energy based beacons and a mobile app," in 2015 IEEE Int. Conf. Consum Electron. (ICCE), 2015, pp. 501–502.
- [22] A. Akinsiku and D. Jadav, "Beasmart: A beacon enabled smarter workplace," in *IEEE/IFIP Netw. Operations Manage. Symp. (NOMS)*, 2016, pp. 1269–1272.
- [23] M. Collotta and G. Pau, "A novel energy management approach for smart homes using bluetooth low energy," *IEEE J. Select. Areas Commun.*, vol. 33, no. 12, pp. 2988–2996, Dec 2015.
- [24] Z. Zhao, J. Fang, G. Q. Huang, and M. Zhang, "ibeacon enabled indoor positioning for warehouse management," in *4th Int. Symp. Comput. Bus. Intell. (ISCBI)*, 2016, pp. 21–26.
- [25] R. Faragher and R. Harle, "Location fingerprinting with bluetooth low energy beacons," *IEEE J. Select. Areas Commun.*, vol. 33, no. 11, pp. 2418–2428, 2015.
- [26] ABIResearch. (2015, December) Abi research predicts wireless connectivity gap to widen as bluetooth enabled device shipments reach 19 billion over the next five years. [Online]. Available: https://www.abiresearch. com/press/abi-research-predicts-wireless-connectivity-gap-wi/
- [27] M. Radhakrishnan, A. Misra, R. K. Balan, and Y. Lee, "Smartphones and ble services: Empirical insights," in *IEEE 12th Int. Conf. Mobile Ad Hoc Sensor Syst. (MASS)*, 2015, pp. 226–234.
- [28] F. Palumbo, P. Barsocchi, S. Chessa, and J. C. Augusto, "A stigmergic approach to indoor localization using bluetooth low energy beacons," in 2015 12th IEEE International Conference on Advanced Video and Signal Based Surveillance (AVSS), Aug 2015, pp. 1–6.
- [29] A. (2017, July) Hong kong Yu. airport set to 'beacon' install system to aid passengers. [Online]. Available: http://www.scmp.com/news/hong-kong/article/1843845/ hong-kong-airport-set-install-beacon-system-aid-passengers
- H. Hamad [30] C. News. (2016,March) international with airport launches mobile ibeacon capabiliapp [Online]. Available: https://dohahamadairport.com/media/ ties. hamad-international-airport-launches-mobile-app-ibeacon-capabilities
- [31] N. Lomas. (2017, May) Gatwick airport now has 2,000 beacons for indoor navigation. [Online]. Available: https://techcrunch.com/2017/05/ 25/gatwick-airport-now-has-2000-beacons-for-indoor-navigation/
- [32] M. Halper. (2017, October) Acuity teams with indoor mapping firm that has big airport presence. [Online]. Available: http://www.ledsmagazine.com/articles/2017/10/ acuity-teams-with-indoor-mapping-firm-that-has-big-airport-presence. html
- [33] A. Ito, H. Hatano, M. Fujii, M. Sato, Y. Watanabe, Y. Hiramatsu, F. Sato, and A. Sasaki, "A trial of navigation system using ble beacon for sightseeing in traditional area of nikko," in 2015 IEEE International Conference on Vehicular Electronics and Safety (ICVES). IEEE, 2015, pp. 170–175.
- [34] P. C. Ng, J. She, and S. Park, "Notify-and-interact: A beacon-smartphone interaction for user engagement in galleries," in 2017 IEEE International Conference on Multimedia and Expo (ICME), July 2017, pp. 1069–1074.
- [35] J. Anderson. (2017, Feb) The icon of modern art puts estimote beacons on display. [Online]. Available: http://blog.estimote.com/post/ 157200820650/the-icon-of-modern-art-puts-estimote-beacons-on
- [36] S. C. Kang. (2017, Nov) Apple daily launches in-app, beacon-based targeting. [Online]. Available: http://www.campaignasia.com/article/ apple-daily-launches-in-app-beacon-based-targeting/441080
- [37] M. Sneps-Sneppe and D. Namiot, "On physical web models," in *IEEE Int. Siberian Conf. Control Commun. (SIBCON)*, 2016, pp. 1–6.

- [38] N. Allurwar, B. Nawale, and S. Patel, "Beacon for proximity target marketing," Int. J. Eng. Comput. Sci, vol. 15, pp. 16359–16364, 2016.
- [39] (2014.Α. Cunningham. July) Explaining continuity: The tech tying ios 8 and os х yosemite together. [Online]. Available: https://arstechnica.com/gadgets/2014/07/ explaining-continuity-the-tech-tying-ios-8-and-os-x-yosemite-together/
- [40] D. Thompson. (2014, Feb) ibeacon auto: Your car is a beacon. [Online]. Available: http://beekn.net/2014/02/ibeacon-auto/
- [41] I. LG Electronics. (2014, Jan) Introducing lg homechat. [Online]. Available: http://www.lghomechat.com/us/index.html
- [42] S. Vigneshwaran, S. Sen, A. Misra, S. Chakraborti, and R. K. Balan, "Using infrastructure-provided context filters for efficient fine-grained activity sensing," in 2015 IEEE International Conference on Pervasive Computing and Communications (PerCom), March 2015, pp. 87–94.
- [43] Y. Kashimoto, T. Morita, M. Fujimoto, Y. Arakawa, H. Suwa, and K. Yasumoto, "Sensing activities and locations of senior citizens toward automatic daycare report generation," in 2017 IEEE 31st International Conference on Advanced Information Networking and Applications (AINA), March 2017, pp. 174–181.
- [44] J. DeCuir, "Introducing bluetooth smart: Part 1: A look at both classic and new technologies." *IEEE Consum. Electron. Mag.*, vol. 3, no. 1, pp. 12–18, 2014.
- [45] C. Gomez, J. Oller, and J. Paradells, "Overview and evaluation of bluetooth low energy: An emerging low-power wireless technology," *Sensors*, vol. 12, no. 9, pp. 11734–11753, 2012. [Online]. Available: http://www.mdpi.com/1424-8220/12/9/11734
- [46] Apple. (2017) ibeacon. [Online]. Available: https://developer.apple.com/ ibeacon/
- [47] Google. (2017, February) Mark up the world using beacons. [Online]. Available: https://developers.google.com/beacons/
- [48] Apple. (2017) Wwdc 2013 videos. [Online]. Available: https: //developer.apple.com/videos/wwdc2013/
- [49] N. Newman, "Apple ibeacon technology briefing," J. Direct, Data and Digital Marketing Practice, vol. 15, no. 3, pp. 222–225, 2014.
- [50] R. (2015,Amadeo. July) Meet google's eddvstone flexible. open source ibeacon fighter. а [Online]. Available: https://arstechnica.com/gadgets/2015/07/ meet-googles-eddystone-a-flexible-open-source-ibeacon-fighter/
- [51] D. Bhattacharya, M. Canul, and S. Knight, "Impact of the physical web and ble beacons," in *Proc. 5th ACM Annu. Conf. Research Inform. Technol.*, 2016, pp. 53–53.
- [52] S. Mittal. (2016, January) ibeacon vs eddystone: Which one works better for your pilot project. [Online]. Available: https: //blog.beaconstac.com/2016/01/ibeacon-vs-eddystone/
- [53] GitHub. (2017) Specification for eddystone, an open beacon format form google. [Online]. Available: https://github.com/google/eddystone
- [54] R.-H. Wu, Y.-H. Lee, H.-W. Tseng, Y.-G. Jan, and M.-H. Chuang, "Study of characteristics of rssi signal," in *IEEE Int. Conf. Ind. Technol.(ICIT)*, 2008, pp. 1–3.
- [55] S. Kajioka, T. Mori, T. Uchiya, I. Takumi, and H. Matsuo, "Experiment of indoor position presumption based on rssi of bluetooth le beacon," in *IEEE 3rd Global Conf. Consum. Electron. (GCCE)*, 2014, pp. 337–339.
- [56] T. S. Rappaport et al., Wireless communications: principles and practice. prentice hall PTR New Jersey, 1996, vol. 2.
- [57] Estimote. (2012-2014) Beacon tech overview. [Online]. Available: http://developer.estimote.com/
- [58] Q. Dong and W. Dargie, "Evaluation of the reliability of rssi for indoor localization," in *IEEE Int. Conf. Wireless Commun. Unusual Confined Areas (ICWCUCA)*, 2012, pp. 1–6.
- [59] R. Faragher and R. Harle, "An analysis of the accuracy of bluetooth low energy for indoor positioning applications," in *Proc. 27th Int. Technical Meeting Satellite Division Inst. Navig. (ION GNSS+ 2014), Tampa, FL,* USA, vol. 812, 2014, p. 2.
- [60] K.-H. Chang, "Bluetooth: a viable solution for iot?[industry perspectives]," *IEEE Wireless Commun.*, vol. 21, no. 6, pp. 6–7, 2014.
- [61] K. E. Jeon, T. Tong, and J. She, "Preliminary design for sustainable ble beacons powered by solar panels," in *IEEE Conf. Comput. Commun. Workshops (INFOCOM WKSHPS)*, 2016, pp. 103–109.
- [62] C. F. Chiasserini and R. R. Rao, "A model for battery pulsed discharge with recovery effect," in WCNC. 1999 IEEE Wireless Communications and Networking Conference (Cat. No.99TH8466), 1999, pp. 636–639 vol.2.
- [63] C. K. Chau, F. Qin, S. Sayed, M. H. Wahab, and Y. Yang, "Harnessing battery recovery effect in wireless sensor networks: Experiments and analysis," *IEEE Journal on Selected Areas in Communications*, vol. 28, no. 7, pp. 1222–1232, September 2010.

- [64] Y. Liu, H. Li, X. Li, J. C. Xue, Y. Xie, and H. Yang, "Self-powered wearable sensor node: Challenges and opportunities (invited)," in *Proceedings* of the 2015 International Conference on Compilers, Architecture and Synthesis for Embedded Systems, ser. CASES '15. Piscataway, NJ, USA: IEEE Press, 2015, pp. 189–189.
- [65] M. Gorlatova, J. Sarik, G. Grebla, M. Cong, I. Kymissis, and G. Zussman, "Movers and shakers: Kinetic energy harvesting for the internet of things," *IEEE Journal on Selected Areas in Communications*, vol. 33, no. 8, pp. 1624–1639, Aug 2015.
- [66] Y. Chen, N. Chiotellis, L. X. Chuo, C. Pfeiffer, Y. Shi, R. G. Dreslinski, A. Grbic, T. Mudge, D. D. Wentzloff, D. Blaauw, and H. S. Kim, "Energyautonomous wireless communication for millimeter-scale internet-ofthings sensor nodes," *IEEE Journal on Selected Areas in Communications*, vol. 34, no. 12, pp. 3962–3977, Dec 2016.
- [67] W.-C. Shih, P. H. Chou, and W.-T. Chen, "A batteryless beacon based on dual ism-band rf harvesting with solar-biasing current," in *Proceedings of the 4th International Workshop on Energy Harvesting and Energy-Neutral Sensing Systems*, ser. ENSsys'16. New York, NY, USA: ACM, 2016, pp. 7–12. [Online]. Available: http://doi.acm.org/10.1145/2996884.2996886
- [68] W. S. Wang, T. O'Donnell, N. Wang, M. Hayes, B. O'Flynn, and C. O'Mathuna, "Design considerations of sub-mw indoor light energy harvesting for wireless sensor systems," ACM Journal on Emerging Technologies in Computing Systems (JETC), vol. 6, no. 2, p. 6, 2010.
- [69] A. Nasiri, S. A. Zabalawi, and G. Mandic, "Indoor power harvesting using photovoltaic cells for low-power applications," *Industrial Electronics*, *IEEE Transactions on*, vol. 56, no. 11, pp. 4502–4509, 2009.
- [70] S. Toulson. (2016) Scan response packet structure. [Online]. Available: https://support.kontakt.io/hc/en-gb/articles/201492522/
- [71] Energizer cr2450. [Online]. Available: http://data.energizer.com/pdfs/ cr2450.pdf
- [72] F. Yin, Y. Zhao, F. Gunnarsson, and F. Gustafsson, "Received-signalstrength threshold optimization using gaussian processes," *IEEE Trans. Signal Process.*, vol. 65, no. 4, April.
- [73] N. Patwari and A. O. H. III, "Using proximity and quantized rss for sensor localization in wireless networks," 2003.
- [74] X. Shen, S. Yang, J. He, and Z. Huang, "Improved localization algorithm based on rssi in low power bluetooth network," in 2nd Int. Conf. Cloud Comput. Internet Things (CCIOT), Oct 2016, pp. 134–137.
- [75] A. Thaljaoui, T. Val, N. Nasri, and D. Brulin, "Ble localization using rssi measurements and iringla," in *IEEE Int. Conf. Ind. Technol. (ICIT)*, March 2015, pp. 2178–2183.
- [76] A. Allan and S. Mistry. (2014, January) Hacking the ces scavenger hunt. [Online]. Available: http://makezine.com/2014/01/ 03/hacking-the-ces-scavenger-hunt/
- [78] P. Misra, S. Raza, V. Rajaraman, J. Warrior, and T. Voigt, "Poster abstract: Security challenges in indoor poster abstract: Security challenges in indoor location sensing using bluetooth le broadcast," in *Poster Demo Session 12th European Conf. Wireless Sensor Netw.*, 2015.
- [79] C. Kolias, L. Copi, F. Zhang, and A. Stavrou, "Breaking ble beacons for fun but mostly profit," in *Proc. ACM 10th European Workshop Syst. Security*, ser. EuroSec'17, New York, NY, USA, 2017, pp. 4:1–4:6.
- [80] R. Pugaliya, J. Chabhadiya, N. Mistry, and A. Prajapati, "Smart shoppe using beacon," in 2017 IEEE International Conference on Smart Technologies and Management for Computing, Communication, Controls, Energy and Materials (ICSTM), Aug 2017, pp. 32–35.
- [81] P. Shende, S. Mehendarge, S. Chougule, P. Kulkarni, and U. Hatwar, "Innovative ideas to improve shopping mall experience over e-commerce websites using beacon technology and data mining algorithms," in 2017 International Conference on Circuit ,Power and Computing Technologies (ICCPCT), April 2017, pp. 1–5.
- [82] Amazon ec2 instance types. [Online]. Available: https://aws.amazon. com/ec2/instance-types/
- [83] P. Somogyi, "Analysis of server-smartphone application communication patterns," Master's thesis, Aalto University, June.
- [84] Apache jmeter. [Online]. Available: http://jmeter.apache.org
- [85] I. Howitt, "Mutual interference between independent bluetooth piconets," *IEEE Trans. Veh. Technol.*, vol. 52, no. 3, pp. 708–718, 2003.
- [86] D. Jenkins and B. Morse, "Application considerations for led light fixtures in process industry production environments," in *IEEE-IAS/PCA Cement Ind. Tech. Conf.*, 2016, pp. 1–6.
- [87] L. Feng, S. Li, H. Li, J. Zhai, Y. Song, L. Jiang, and D. Zhu, "Superhydrophobic surface of aligned polyacrylonitrile nanofibers," *Angewandte Chemie*, vol. 114, no. 7, pp. 1269–1271, 2002.

- [88] A. Yuji and T. Yachi, "A novel photovoltaic module assembled threedimensional," in 2010 35th IEEE Photovoltaic Specialists Conference, June 2010, pp. 002 811–002 816.
- [89] M. Bernardi, N. Ferralis, J. H. Wan, R. Villalon, and J. C. Grossman, "Solar energy generation in three dimensions," *Energy & Environmental Science*, vol. 5, no. 5, pp. 6880–6884, 2012.
- [90] K. Liu, J. Du, J. Wu, and L. Jiang, "Superhydrophobic gecko feet with high adhesive forces towards water and their bio-inspired materials," *Nanoscale*, vol. 4, pp. 768–772, 2012.
- [91] H. Cho, J. Ji, Z. Chen, H. Park, and L. Wonsuk, "Accurate distance estimation between things: A self-correcting approach," *Open J. Internet Things (OJIOT)*, 2015, vol.1, issue 2, pp. 19-27.
- [92] C. Liu, P. Zhao, K. Bian, T. Zhao, and Y. Wei, "The detection of physical attacks against ibeacon transmitters," in *IEEE/ACM 24th Int. Symp. Quality of Service (IWQoS)*, 2016, pp. 1–10.